Author's response to reviews

Title: The supply of General Practitioners across local areas: accounting for spatial heterogeneity

Authors:

Michelle McIsaac (m.mcisaac@outlook.com)
Anthony Scott (a.scott@unimelb.edu.au)
Guyonne Kalb (g.kalb@unimelb.edu.au)

Version: 2 Date: 23 May 2015

Author's response to reviews:

Revisions made to the manuscript:

1. Line numbers have been added to the manuscript.
2. Page breaks have been removed.
3. No primary data collection was used in this paper and the paper pertains to geographical areas, therefore an ethics statement was not added.
4. Competing interests declaration was added after the conclusion.